### The WTO and Direct Taxation

by

Michael Daly\*

World Trade Organization Geneva, Switzerland

June 2005

\*This paper was prepared partly for the High Level Scientific Conference, "WTO and Direct Taxation", organized by the Department of Austrian and International Tax Law at the University of Vienna and held in Rust on 8-11 July 2004. The author has benefited from discussions with Reuven Avi-Yonah, Michael Lang, Albert Rädler and other participants in the Conference. He is especially indebted to Jesse Kreier, Michael Lennard and Bruce Wilson for their many detailed comments. The opinions expressed are solely those of the author, and, as such, should not be attributed to the WTO Secretariat.

This paper is only available in English – Price CHF 20.-

To order, please contact:

WTO Publications Centre William Rappard 154 rue de Lausanne CH-1211 Geneva Switzerland

Tel: (41 22) 739 52 08 Fax: (41 22) 739 57 92 Website: www.wto.org

E-mail: publications@wto.org

ISSN 1726-9466 ISBN 92-870-3327-7 Printed by the WTO Secretariat VI-2005

Keywords: direct taxation, WTO rules, foreign sales corporation (FSC), trade disputes. JEL classification code: F13, H25, K34

© World Trade Organization, 2005. Reproduction of material contained in this document may be made only with written permission of the WTO Publications Manager.

With written permission of the WTO Publications Manager, reproduction and use of the material contained in this document for non-commercial educational and training purposes is encouraged.

### **ABSTRACT**

The widely-followed ruling by the WTO's Dispute Settlement Body against the United States concerning the latter's FSC/ETI scheme, which led to the largest retaliation award ever authorized in a dispute at the WTO, confirmed (if there were ever any doubt) that, generally speaking, direct taxes, like indirect taxes (including tariffs), are subject to the multilateral rules of the WTO, notwithstanding efforts by tax authorities to secure specific exemptions for certain direct tax measures in these agreements. This ruling reconfirmed the traditional distinction under international trade rules between direct and indirect taxes, particularly with respect to how such taxes should be treated under the subsidy and border tax adjustment rules of the WTO. It prompted the US Congress finally to pass legislation in late 2004 to repeal the FSC/ETI scheme as part of a larger overhaul of the US corporate tax system. The most recent disputes between the United States and the European Communities over assistance to large civil aircraft (allegedly amounting in each case to even more than the FSC/ETI) also encompass direct tax measures.

It would be not be surprising if other WTO-inconsistent direct tax measures were identified in the future, leading to further disputes among WTO Members. Multilateral WTO rules, which are agreed by consensus, can therefore be expected to continue to be an important factor in how Members' shape their tax policies as they will undoubtedly want to avoid having their tax policies successfully challenged in the WTO.

### TABLE OF CONTENTS

I.	INTRODUCTION	1
II.	TAXATION IN WTO AGREEMENTS	4
III.	DISPUTES AT THE WTO CONCERNING DIRECT TAXATION	9
IV.	SOME DIRECT TAX MEASURES IDENTIFIED IN TRADE POLICY REVIEWS	13
V.	RATIONALE AND PRINCIPLES UNDERLYING WTO RULES AND INTERNATIONAL TAXATION	16
VI.	INCIDENCE OF DIRECT TAXES	22
VII.	CONCLUSIONS	25
VIII.	REFERENCES	26

### I. INTRODUCTION

International rules concerning measures that affect trade and those regarding direct taxation appear to have broadly similar goals, namely the removal of obstacles to the cross-border movement of goods, services, capital, labour and technology. Whereas international trade rules pertain largely to the movement of goods and services, those with respect to direct taxation mainly involve factors of production, especially capital. Notwithstanding their broadly similar goals, and the apparent analytical equivalence between cross-border flows of products and flows of factors<sup>1</sup>, the two sets of rules have, in general, evolved quite separately.2 Binding rules concerning tariffs and non-tariff barriers (NTBs) to trade have been negotiated multilaterally under the auspices of the General Agreement on Tariffs and Trade (GATT) and its successor, the World Trade Organization (WTO). By contrast, rules concerning direct taxation have been either decided unilaterally, as in the case of the methods used to relieve international double taxation<sup>3</sup>, or negotiated bilaterally, in the case of withholding tax rates on cross-border income flows. These bilateral negotiations have been based to a large extent on the OECD Model Tax Convention on Income and on Capital ("the Model"), which evolved from working groups established by the International Chamber of Commerce and the League of Nations in the 1920s.

-

GATT/WTO rules recognize that internal taxation, including direct tax measures, can, like other laws and regulations, have economic effects that are similar, if not equivalent, to tariff and non-tariff border measures as well as production and export subsidies. They can also be used to deter or encourage international flows of capital and labour. With tariffs declining as a consequence of successive rounds of multilateral trade negotiations during the past 50 years or so under the General Agreement on Tariffs and Trade (GATT<sup>4</sup>), attention has focused increasingly on non-tariff measures, including internal taxes. This has raised concerns that internal tax measures may vitiate the effects of tariff reductions on international trade in goods, as it has long been known that such measures can have effects equivalent to tariffs.<sup>5</sup> Moreover, with trade in services accounting for around three-quarters of production and employment in industrialized countries, it is recognized that non-tariff restrictions, particularly internal regulations, can be important obstacles to the cross-border provision of services, which is more likely to require providers of such services to be located close to consumers, although services may also be embodied in goods. Consequently, obstacles to foreign direct investment (FDI) impede the establishment of foreign firms that provide these services; likewise, measures to encourage FDI facilitate trade in services.<sup>6</sup> Liberalization of trade in services and of FDI thus often go hand in hand.

<sup>&</sup>lt;sup>1</sup> This equivalence is a long-recognized implication of mainstream international trade theory (the Heckscher-Ohlin model) that product flows and factor flows are substitutes for each other in the sense that a similar equilibrium world price ratio will prevail with either products freely mobile and with factors immobile, or with products immobile but with factors mobile (Mundell, 1957). There is also a close relationship between a country's trade deficit (or surplus) and its net foreign investment; a country whose imports exceed exports is likely to finance the resulting trade deficit by promising to pay foreigners in the future, thereby increasing inbound foreign investment (Auerbach and Kotlikoff, 1995). Hence, trade policies affecting imports will have similar effects to tax policies affecting international investment.

<sup>&</sup>lt;sup>2</sup> See Warren (2000) for a more detailed discussion of this evolution.

<sup>&</sup>lt;sup>3</sup> International double taxation has long been recognized as an impediment to international trade and investment It arises most obviously if cross-border investments are taxed both in the country where the investment is made (the "source" country) and the country where the investor resides (the "residence" country).

<sup>&</sup>lt;sup>4</sup> The overall import-weighted tariff average for manufactured products in industrialized countries is currently 6.3 per cent and will fall to 3.9 per cent, once commitments made under the Uruguay Round are fully implemented.

See Slemrod (1995), for example. Bond and Guisinger (1985) found that Ireland's replacement of tariffs with tax as well as direct incentives for investment after joining the European Community left Irish industry in roughly the same position with regard to the domestic protection of labour as before. This finding is hardly surprising given that the economic and revenue effects of an import tariff can be replicated by the combination of a destination-based consumption tax plus either a subsidy or a tax exemption to domestic production at equal rates.

<sup>&</sup>lt;sup>6</sup> Commercial presence (mode 3), involving FDI, is estimated to account for around 56 per cent of total world trade in services.

In recognition of its potentially important effects on international trade and investment flows, taxation is coming under increased scrutiny at the WTO.<sup>7</sup> This recognition is reflected in several of the multilateral agreements reached at the conclusion of the Uruguay Round in December 1993. These agreements include those on Subsidies and Countervailing Measures (SCM), Agriculture, Trade-Related Investment Measures (TRIMs), and services (GATS), which encompass direct as well as indirect taxation. These agreements reflect the growing realization on the part of national governments that multilateral rules need to play an increasingly important role in regulating the use of tax measures, especially where these measures affect the international movement of goods, services, capital, persons and technology. The expansion of WTO rules concerning trade and investment has increased the potential for conflict between these rules and tax laws and hence the scope for the WTO to encroach on Members' freedom to decide their own internal tax policies. Not surprisingly, therefore, disputes concerning taxation are becoming more frequent at the WTO, implying inconsistency between Members' tax laws and WTO agreements.

Hitherto, most of these disputes have involved indirect taxes, including excise taxes on alcoholic beverages (in Japan, Korea and Chile) and periodicals (in Canada), sales taxes on automobiles (in Indonesia), and VAT on integrated circuits (in China). However, in a recent high-profile and contentious case involving direct taxation, initiated by the European Communities (EC), a WTO Panel, upheld by the Appellate Body, ruled that the United States' "foreign sales corporation" (FSC) scheme (and its replacement, the Extraterritorial Income Exclusion Act (ETI)) constituted a prohibited export subsidy and, as such, violated the Agreement on Subsidies and Countervailing Measures as well as that on Agriculture. The amount of retaliation authorized when the United States failed to implement the ruling within the time period specified by the Dispute Settlement Body (DSB) was some US\$4 billion per annum, by far the largest amount ever awarded in a WTO dispute. This ruling will undoubtedly be of interest as Members formulate

<sup>7</sup> Taxation was also a major issue during negotiations at the OECD regarding the Multilateral Agreement on Investment (Daly, 1997).

their domestic tax policies in the future, especially where such policies differentiate between exportoriented and domestically-oriented sectors, and may even move Members to reformulate certain of these policies, as was the case with the US Furthermore, the dispute over the FSC prompted the US to initiate six other cases with the EC over certain income tax measures in Member States, although these cases have never gone beyond the consultation phase. More recently, another major dispute between the US and the EC concerning government assistance to Airbus and Boeing, respectively, has gone to the WTO, with (at the time of writing) both Members requesting the establishment of separate panels. In the case of the EC's complaint over government assistance to Boeing, it is alleged that such assistance involves tax incentives provided not just by the US Federal Government, notably the FSC/ETI scheme, of which Boeing was one of the main beneficiaries, but also tax incentives provided by some state and local governments, including those amounting to US\$3 billion or so from the State of Washington, where much of Boeing's main manufacturing facility is located.

The WTO's interest in taxation is not confined to ensuring Members' compliance with the rules of its various agreements, however. Irrespective of whether they comply with WTO rules, Members' tax policies, practices and measures are being monitored much more closely under the WTO Trade Policy Review Mechanism (TPRM). Such monitoring throws light on how tax measures not yet covered by WTO rules may nonetheless have economic effects equivalent to (or more important than) tax or non-tax measures prohibited by existing WTO rules. By identifying new issues, the TPRM may pave the way for the adoption of further multilateral rules concerning taxation. On the other hand, as these measures tend to be more detrimental to the economy of the country imposing them than to the economies of its trading partners, a case can be made for removing or modifying such measures unilaterally, regardless of whether they are permitted by WTO rules.8

<sup>&</sup>lt;sup>8</sup> Historically, unilateral liberalization, which is usually linked to a broader program of domestic reform, has accounted for most of the reductions in border protection. Most comprehensive trade reforms among large countries (Argentina, Brazil, China and India in the early 1990s) were primarily unilateral reforms that were undertaken to increase competition, and thus productivity, in the domestic economy.

The rest of this paper is organized as follows. The next section provides an overview of the extent to which present WTO rules apply to taxation, particularly direct taxation. Section 3 contains a summary of disputes concerning income tax measures that have arisen at the WTO since its establishment, with special attention to the protracted dispute between the EC and the US over the latter's FSC/ETI schemes. Section 4 focuses on some issues regarding direct taxation that have been identified in recent trade policy reviews; the tax policies and measures involved do not necessarily contravene WTO rules.

Section 5 provides a comparison of the rationale and principles underlying WTO rules and those underlying international taxation. Section 6 reviews some recent empirical evidence regarding the incidence of direct taxes, which has a bearing on existing WTO rules concerning border tax adjustment in respect of exports and also on the assessment of damages in respect of tax measures found to contravene these rules. Some concluding remarks are found in Section 7.

### II. TAXATION IN WTO AGREEMENTS

A. THE GENERAL AGREEMENT ON TARIFFS AND TRADE (GATT), 1994

A WTO Member is legally obliged not to raise border taxes in the form of tariffs above the specified rates agreed in GATT negotiations and incorporated into its schedule of concessions. The tariff rates so agreed are known as "bound" rates. Their purpose is to provide greater commercial certainty through the establishment of a ceiling on tariffs that cannot be breached without an offer of compensation to affected trading partners. Bound tariff rates are often higher than "applied" rates.

The cornerstone of the GATT, as well as other WTO agreements, is non-discrimination; it has two aspects, namely most-favored-nation (MFN) and national treatment (NT). The MFN principle, which is embodied in Article I (*General Most-Favoured-Nation Treatment*) of the GATT, stipulates that countries should not discriminate between trading partners' goods; that is,

" ... any advantage, favor, privilege or immunity granted by any contracting party to any product originating in or destined for any other country shall be accorded immediately and unconditionally to the like product originating in or destined for the territories of all other contracting parties."

In other words, concessions accorded to one country's goods should be granted to those of *all* countries.<sup>9</sup> The MFN principle thus ensures import neutrality as far as goods are concerned.

The principle of NT is embodied in Article III (National Treatment on Internal Taxation and Regulation) of the GATT. In particular, Article III(1) states that

"(T)he contracting parties recognize that internal taxes and other internal charges,

9 Article XXIV permits departures from the MFN principle in respect of regional and bilateral free trade agreements, provided these preferential (and therefore discriminatory) agreements cover substantially all trade between the parties. More than 200 such agreements are currently in force even

though no determination has yet been made by the WTO as to whether any of them is actually in accordance with Article XXIV.

and laws, regulations and requirements affecting the internal sale, offering for sale, purchase, transportation, distribution or use of products, .... should not be applied to imported or domestic products so as to afford protection to domestic production."

### Article III(2) requires that

" ... products of the territory of any contracting party imported into the territory of any other contracting party shall not be subject, directly or indirectly, to internal taxes or other internal charges of any kind in excess of those applied, directly or indirectly, to like domestic products. Moreover, no contracting party shall otherwise apply internal taxes or other internal charges to imported or domestic products in a manner contrary to the principles set forth in paragraph 1."

### In addition, Article III(4) states that

" ... products of the territory of any contracting party imported into the territory of any other contracting party shall be accorded treatment no less favourable than that accorded to like products of national origin in respect of laws, regulations and requirements affecting their internal sale, offering for sale, purchase, transportation, distribution or use."

In other words, imported goods must be treated the same as or no less favourably than domestically-produced goods so as to ensure that discriminatory internal taxes are not used as substitutes for tariffs.

As those GATT rules concerning import barriers pertain to goods only, it would appear that these rules relate more to indirect taxes (border and internal) than to direct taxes.<sup>10</sup> The lack of

<sup>&</sup>lt;sup>10</sup> See Lennard (2004). According to Jackson (1969, p. 297), for example, "(T)he interpretation was made several times that income taxes did not fall within the language like that of Article III, paragraph 2, because that language 'is concerned solely with internal taxes on goods'". But given the apparent equivalence between flows of products and factors and indirect taxes are levied on the former and direct taxes on the latter, the two types of taxes can have equivalent effects.

disputes, until recently, was viewed by some as providing support for the view that there was little or no scope for coverage by the GATT 1947 of direct taxes because such taxes did not relate sufficiently to products. However, the recent Panel Report by the WTO on the FSC/ETI (discussed below) makes abundantly clear that such taxes do indeed fall within the scope of Article III of the GATT. More specifically, the Panel stated that

"(I)n this connection, we can see no specification or limitation in the text of Article III:4 concerning the type of advantage linked to the measure under examination under Article III:4 of the GATT 1994. Thus, nothing in the plain language of the provision specifically excludes requirements conditioning access to income tax measures from the scope of application of Article III".<sup>11</sup>

The Panel went on to conclude that

"Article III:4 of the GATT 1994 applies to measures conditioning access to income tax advantages in respect of certain products."

As regards exported goods, in a 1960 draft declaration giving effect to the provisions of Article XVI:4 (*Subsidies*) of the GATT, the "remission, calculated in relation to exports, of direct taxes or social welfare charges on industrial or commercial enterprises" are considered as export subsidies, whereas tariff or consumption tax refunds on exports are not. Similar language is found in Annex I (e) of the SCM Agreement (see below). Furthermore, Ad Article XVI of the GATT specifies that

"(T)he exemption of an exported product from duties or taxes borne by the like product when destined for domestic consumption, or remission of such duties or taxes in amounts not in excess of those which have accrued, shall not be deemed a subsidy."

Panel Report, United States – Tax Treatment for "Foreign Sales Corporations", Recourse to Article 21.5 of the DSU by the European Communities, WT/DS108/RW, adopted 29 January 2002, as modified by the Appellate Body Report, WT/DS108/

AB/RW, paragraphs 8.142-8.143.

Also noteworthy with regard to direct taxation is an "Understanding" reached by the GATT Council in 1981 as a result of four panel reports issued in connection with disputes between the United States and the European Community over direct taxation. These disputes involved a complaint lodged in February 1972 by the EC against the United States' Domestic International Sales Corporation (DISC) scheme, which exempted so-called DISC income from corporate income tax and allowed partial deferment of tax on that income received by shareholders, and the related US complaints against the "territorial" income tax systems of Belgium. France and the Netherlands. 12 The GATT panel, whose report was issued in November 1976, found elements of subsidy in both the United States' DISC and "territorial" tax systems.

Following the issuance of the panel reports, the next step in the dispute-resolution process would normally have been for the GATT Council to consider whether or not to adopt these reports. However, the United States and the EC had reached an impasse. The United States was willing to accept the finding against the DISC, but only if the findings against the three "territorial" systems were also adopted; the three European countries were only willing to adopt the finding against the DISC. As a result, a GATT Council decision remained in abeyance for nearly five years. The hiatus was partly due to the increased opposition to the DISC within the United States itself. In addition, the negotiation of a new GATT Subsidies Code during the Tokyo Round appeared to point to a resolution of the dispute. The Code explicitly included tax deferrals as subsidies, casting doubt on the DISC's legality, while affirming the importance of using arm's-length pricing in intra-firm transfers, a key element of the United States' complaint against "territorial" systems. At the same time, the Subsidies Code appeared to countenance the legality of "territorial" systems in general.

In December 1981, the four panel reports were finally adopted by the GATT Council, subject to an Understanding. The Understanding, which later assumed an important role in the United States' defence of the Foreign Sales Corporation (FSC) at the WTO, involved agreement on three

5

<sup>&</sup>lt;sup>12</sup> See Brumbaugh (2004).

points. First, countries need not tax economic processes occurring outside their territory. Notwithstanding the panel reports, "territorial" tax systems did not, in other words, generally contravene the GATT. Second, arms-length pricing should be followed in allocating income among related firms. Third, the GATT does not prohibit measures designed to alleviated double-taxation of foreign-source income.

However, the 1981 Understanding and adoption of the reports did not end the controversy. Indeed, the meaning of the Understanding itself shortly became a bone of contention, and in 1982 a debate occurred in the GATT Council over how the Understanding applied to the DISC scheme. The EC continued to argue that the DISC was an illegal export subsidy, while the United States maintained that it was merely approximating the effect of the "territorial" tax system commonly used by European countries. In 1984, however, the United States did replace the DISC with the Foreign Sales Corporation (FSC) scheme, which remained unchallenged until 1998.

# B. AGREEMENT ON TRADE-RELATED INVESTMENT MEASURES (TRIMS)

Not only do governments use tax as well as non-tax incentives to attract foreign investment, they may also impose conditions to ensure that the investments accord with certain national priorities. Such conditions include, *inter alia*: local content provisions, which require the investor to utilize a certain amount of local (instead of imported) inputs in production; and export performance requirements that compel the investor to export a certain proportion of its output. Such conditions, which can distort trade, just like import tariffs (or quantitative restrictions) and export subsidies, are known as trade-related investment measures (TRIMs).

The TRIMs Agreement applies to investment measures related to trade in goods only. It prohibits WTO Members from applying any TRIM that is inconsistent with Articles III (National Treatment) and XI (General Elimination of Quantitative Restrictions) of the GATT. The illustrative list of prohibited TRIMs contains four categories of measures: (a) benefits that are conditional upon local content requirements; (b) the conditioning of a firm's ability to import on its export performance; (c) foreign exchange

balancing requirements or restrictions; and (d) domestic sales requirements involving restrictions on exports. Prohibited TRIMs thus include not only mandatory measures but also those "with which compliance is necessary to obtain an advantage". Although the term "advantage" is not defined explicitly in the Agreement, it is understood to encompass all types of advantages, including tax relief. The Agreement is limited in scope, however. It is noteworthy, for example, that it does not prevent countries from attaching export performance requirements to tax or nontax incentives for investment. Nor does it prevent them from requiring that a minimum percentage of equity be held by local investors or that the foreign investor must bring in the most up-to-date technology or must conduct a certain amount or type of R&D locally.

# C. AGREEMENT ON SUBSIDIES AND COUNTERVAILING MEASURES (SCM)

The SCM Agreement regulates the provision by WTO Members of subsidies with respect to goods and the actions that can be taken against such subsidies by a WTO Member (the application of the SCM to agricultural products is, however, limited in some respects by the Agreement on Agriculture). Article 1 (*Definition of a Subsidy*) of the Agreement defines a subsidy as a financial contribution by a Member of the WTO or by any public body within the territory of a Member which confers a benefit. Such contributions include revenue forgone as a consequence of tax incentives.<sup>13</sup>

The SCM Agreement divides subsidies into prohibited and permissible subsidies. Export subsidies (that is, those contingent on export performance) and subsidies contingent on the use of domestic over imported goods are prohibited. The SCM Agreement distinguishes between two categories of permissible subsidies: those that are

6

<sup>&</sup>lt;sup>13</sup> They also include direct transfers of funds, actual (e.g., grants, loans, equity) or potential (e.g., loan guarantees), provision of goods or services other than general infrastructure, purchases of goods and price support. In accordance with Article XVI of the GATT, however, the definition does not include the refund or remission of duties or indirect taxes on goods upon exportation.

actionable and those that are non-actionable.14 Annex I of the SCM Agreement contains an illustrative list of export subsidies. Item (e) of the list refers to measures involving "full or partial exemption, remission, or deferral specifically related to exports, of direct taxes". 15 Footnote 59 makes clear that item (e) is not intended to prevent Members from taking measures to avoid double taxation of foreign-source income. Further, item (f) treats as an export subsidy "...special deductions directly related to exports or export performance, over and above those granted in respect to production for domestic consumption, in the calculation of the base on which direct taxes are charged". By contrast, and in accordance with Article XVI of the GATT, the SCM Agreement provides that the exemption or remission of an exported product from indirect taxes or import duties upon exportation does not constitute a subsidy.

### D. AGREEMENT ON AGRICULTURE (AoA)

Under Article 1 (Definition of Terms) of the Agreement on Agriculture (AoA), "budgetary outlays" or "outlays" include revenue forgone, so that tax measures are covered by the Agreement insofar as they constitute export subsidies. The disciplines on subsidies agreed for the agriculture sector are quite different from those found in the SCM Agreement in at least one important respect. While the AoA has established rules concerning the acceptability of various subsidization practices – "green box" measures are acceptable,

"amber box" measures are not<sup>16</sup> – it also involves commitments to <u>reduce</u> aggregate levels of support. Hence, subsidy disciplines are designed in accordance with commitments to a progressive reduction in levels of subsidization.

# E. General Agreement on Trade in Services (GATS)

In contrast to the GATT, SCM and TRIMs agreements, which deal solely with trade in goods, the GATS applies to trade in services. The GATS constitutes a first attempt to subject the provision of services to international trading rules embodying the principles of MFN and national treatment.<sup>17</sup> The concept of NT in the GATS is of particular relevance because it involves non-discrimination on the basis of the origin, not only of the services, but of the service suppliers. Hence, FDI is covered by the GATS insofar as it involves a commercial presence for the delivery of services. Under Article XIV (General Exceptions), to the extent that countries have made NT (Article XVII) commitments in their schedules, these commitments apply to tax measures (including tax incentives), except where such measures are aimed at ensuring "the equitable or effective imposition or collection of direct taxes in respect of services or service suppliers of other Members". This implies that direct tax measures would generally be covered by Article XVII<sup>18</sup>; otherwise Uruguay Round negotiators would not have deemed it necessary to create such an explicit exception when drafting the relevant provisions of the GATS. Also, tax measures that depart from the MFN treatment obligation (Article II) are permitted if they are the result of a bilateral agreement on the avoidance of double taxation (or similar binding provisions in other international agreements).

<sup>&</sup>lt;sup>14</sup> Broadly speaking, this distinction is based on the concept of "specificity". Specific subsidies are actionable insofar as they have "adverse effects" on the interests of other Members. Although other subsidies – in particular, certain subsidies for environmental, research and development and regional development – were for a time non-actionable under Article 8 of the SCM, that provision expired at the end of 1999.

<sup>&</sup>lt;sup>15</sup> For the purpose of the SCM Agreement, "direct taxes" refer to taxes on wages, profits, interest, rents, royalties, and all other forms of income, and taxes on the ownership of real property (see footnote 58 of the SCM Agreement). "Indirect taxes" mean sales, excise, turnover, value added, franchise, stamp, transfer, inventory and equipment taxes, border taxes and all taxes other than direct taxes and import charges.

<sup>&</sup>quot;Green" subsidies are those that have "no, or minimal, tradedistorting effects or effects on production" and do not have the "effect of providing price support to producers" and are thus exempt from reduction commitments. Such commitments do apply to "amber" subsidies, which include certain direct payments under production-limiting programmes (sometimes dubbed "blue" box measures).

<sup>&</sup>lt;sup>17</sup> Unlike in the GATT, NT under the GATS is not a general commitment; it applies only to scheduled sectors.

 $<sup>^{18}</sup>$  The meaning of "equitable or effective" is spelled out in footnote 6 to article XIV(d).

### F. TRADE POLICY REVIEW MECHANISM (TPRM)

The above provisions of WTO agreements are not the only ones to exert disciplines on the use of tax measures. The notification obligations contained in these agreements also enhance transparency.<sup>19</sup> Transparency is further enhanced by the Trade Policy Review Mechanism (TPRM). The purpose of the TPRM is to contribute to improved adherence by all Members to rules, disciplines and commitments made under the WTO agreements, and hence to the smoother functioning of the multilateral trading system, by achieving greater transparency in, and understanding of, Members trade policies and practices, including tax measures. In the case of taxation, transparency entails four key elements: (a) a description of the nature of tax measures; (b) their rationale or objectives; (c) their cost (or benefits) in terms of revenue forgone (or taxes collected)<sup>20</sup>; and (d) an economic evaluation of the effectiveness of individual tax measures (relative to alternative measures) in achieving their given objectives. Accordingly, the mechanism enables the regular collective appreciation and evaluation

of a full range of individual Members' trade policies and practices and their impact on the functioning of the multilateral trading system. The TPRM is not, however, intended to serve as a basis for the enforcement of specific obligations under the Agreements, or for dispute settlement purposes. Nor is it intended to impose new policy commitments on Members. Nevertheless, it does permit the evaluation of trade and trade-related policies and measures, even though they may not necessarily contravene, or indeed be subject to, WTO obligations.

The mechanism involves periodic<sup>21</sup> reviews by the Trade Policy Review Body of each Member's trade and trade-related policies, practices and measures based mainly on a report drawn up by the WTO Secretariat on its own responsibility and a report supplied by the Member (or Members) under review. Both reports are published after the review

8

<sup>&</sup>lt;sup>19</sup> Under the SCM Agreement's notification (or transparency) mechanism, for example, Members are obliged to notify the WTO of all existing subsidies as defined in Article 1, that are specific within the meaning of Article 2, whether prohibited, actionable or non-actionable. (Members were also required to notify the WTO of any prohibited subsidies in existence when the Agreement was signed.) In accordance with Article 25, this notification must be sufficient to enable other Members to understand how the subsidy operates and to evaluate its effects on international trade in goods. It should include budgetary information, but not confidential business information. Likewise, under the TRIMs Agreement, Members are required to notify the WTO of all TRIMs that are not in conformity with the provisions of the Agreement.

<sup>&</sup>lt;sup>20</sup> Several WTO Members publish tax expenditure accounts which show the amounts of tax revenues forgone as a consequence of various forms of tax relief. Some Members also publish studies evaluating the effectiveness of such measures.

<sup>&</sup>lt;sup>21</sup> The US, EC, Japan and China are reviewed every two years; the next 16 largest traders are reviewed every four years; other Members are reviewed every six years, although a longer period may be allowed for least-developed country Members.

### III. DISPUTES AT THE WTO CONCERNING DIRECT TAXATION

The expansion of WTO rules concerning international trade and investment has increased the potential for conflict between these rules and Members' tax laws and thus for disputes between Members. Disputes at the WTO imply inconsistency between WTO rules and domestic as well as international tax laws. Hitherto, out of a total 330 disputes initiated at the WTO (with a request for consultation), 34 have been over taxation, 24 of which entailed indirect taxes and the other 10 direct taxes. By and large, disputes over indirect taxes have involved alleged differential treatment of imported products (including periodicals, beverages, cigarettes, automobiles and integrated circuits) in relation to like domestic products, in violation of GATT Article III (National Treatment on Internal Taxation and Regulation). However, there have also been several disputes involving direct taxation, the most notable of which involved the US FSC/ETI schemes. These disputes are outlined below, with special attention focused on the FSC/ETI schemes because of their potentially far-reaching implications.

#### A. TURKEY'S TAXATION OF FOREIGN FILMS

In June 1996, the US requested consultations with Turkey regarding the latter's taxation of revenues generated from the showing of foreign films, which it alleged violated GATT Article III concerning NT. The Dispute Settlement Body (DSB) established a panel in February 1997 to hear the case. In July 1997, however, both parties notified the WTO of a mutually agreed solution.

# B. UNITED STATES' FOREIGN SALES CORPORATION (FSC) AND EXTRATERRITORIAL INCOME (ETI) SCHEMES<sup>22</sup>

The dispute between the United States and the European Community that resulted in the "1981 Understanding" re-surfaced in November 1997, when the EC formally challenged the US over the DISC's successor, the FSC scheme, which was enacted in 1984. The FSC, like the DISC and subsequent ETI, was intended to offset the

<sup>22</sup> A more detailed history and description of the DISC/FSC/ ETI measures and resulting disputes between the EC and US can be found in Brumbaugh (2004). perceived tax disadvantage encountered by US producers in respect of their exports.<sup>23</sup> When consultations failed to resolve the dispute, the EC requested the establishment of a dispute settlement panel at the WTO and such a panel was formed in September 1998.

The EC argued that the FSC constituted a violation of the Agreement on Subsidies and Countervailing Measures (SCM) and the AoA, which entered into force in 1995. Specifically, the EC complained that the FSC violated Article 3.1(a) and 3.1 (b) of the SCM, which prohibit provision of subsidies that are contingent upon export performance or upon the use of domestic over imported goods, as well as Articles 3, 8, 9 and 10 of the AoA. In general, the EC argued that the FSC scheme's special tax exemptions for export income together with "administrative pricing" rules conferred a benefit on US exporters and therefore constituted an export subsidy in violation of WTO rules.<sup>24</sup>

For its part, the US claimed that the FSC was analogous to "territorial" taxation, which was consistent with WTO rules. More specifically, the US maintained that the "1981 Understanding" and the SCM provided that countries need not tax income from "foreign economic processes," the basis for the acceptability of "territorial" taxation. The US then argued that FSC's various foreign economic presence and management requirements linked the tax benefit to foreign economic

<sup>&</sup>lt;sup>23</sup> This disadvantage, it was argued, involved two elements. The first element concerned the fact that whereas some European countries generally tax only income earned domestically (the so-called "territorial" system of direct taxation), the US generally taxes world-wide income of its residents (the so-called "world-wide" system of direct taxation). Secondly, whereas US exports to the EU are subject to the EU value-added tax (VAT), EU exports to the US are not because the EU exempts its own imports and the US does not levy such a tax. This is argued by some to give a distinct advantage to EU exports to the US not enjoyed by US exports to the EU. Apparently, the combination of specified exemptions and pricing rules embodied in the FSC amounted to a total tax exemption of between 15 and 30 per cent of income from exports.

<sup>&</sup>lt;sup>24</sup> The FSC, like the DISC, provisions also contained "administrative" income allocation rules that assigned more income to the tax-favoured FSC subsidiary than could be allocated under "arm's length" pricing method, although firms could use the latter method if they wished.

processes. The US further maintained that the SCM Agreement gives countries considerable scope for setting required intra-firm pricing methods, and FSC's "administrative pricing" rules merely identified the portion of income attributable to foreign economic processes.

The WTO Panel, whose Report was issued in October 1999, concluded that the "carve-out" of foreign-source income attributable to exports from the "world-wide" income tax system allowed by the FSC did indeed constitute a prohibited subsidy contingent on exporting, and thus violated Article 3.1(a) of the SCM.<sup>25</sup> The Panel also found that the US had acted inconsistently with its obligations under Article 3.3 of the AoA (and consequently with its obligations under Article 8 of that Agreement).

In reaching its conclusion, the Panel rejected the US analogy between FSC and "territorial" taxation. Although the Panel accepted that countries need not tax income from foreign economic processes, whether a provision forgives taxes "otherwise due", and therefore constitutes a subsidy, depends on how the provision in question compares to a country's own general method of taxation. In short, because the FSC carved out an exception from the way the US normally taxed income from exports, it was, the Panel concluded, an export subsidy.

The US appealed the Panel's decision almost immediately, arguing again that the WTO rules hold that a country need not tax income from foreign economic processes and that FSC was therefore permissible. However, on 22 February 2000, the WTO Appellate Body ruled as follows:

- it upheld the Panel's finding that the FSC measure involved subsidies "contingent upon export performance" and therefore constituted a prohibited subsidy under SCM Article 3.1(a);
- it reversed the Panel's finding that the FSC measure involved "the provision of subsidies to reduce the costs of marketing exports" of agricultural products under Article 9.1(d)

of the AoA and, in consequence, reversed the Panel's findings that the US had acted inconsistently with its obligations under Article 3.3 of AoA:

- it found that the US acted inconsistently with its obligations under Articles 10.1 and 8 of the AoA by applying export subsidies, through the FSC measure, in a manner which results in, or threatens to lead to, circumvention of its export subsidy commitments with respect to agricultural products; and
- it also emphasized that it was not ruling that a Member must choose one kind of tax system over another so as to be consistent with that Member's WTO obligations, or on the relative merits of the "worldwide" and "territorial" systems, while rejecting US arguments invoking footnote 59 of the SCM, which covers measures to relieve international double taxation.

More broadly, the Appellate Body concluded that, having decided to tax foreign source income in general, the US provided a subsidy by carving out an exception to that treatment, thus once again rejecting the analogy between FSC and "territorial" taxation. The DSB adopted the Appellate Body Report and the Panel Report, as modified by the Appellate Body Report, at its meeting in March 2000.

In the face of the Appellate Body's decision, the US was confronted with the following dilemma: either make some adjustments to the FSC scheme's design to make it comply with WTO rules, or undertake broad reform of its income tax system - with some prominent policy-makers suggesting that the US adopt a broad-based consumption tax and others advocating adoption of the "territorial" tax system. In the event, the US opted for the narrower approach, enacting the ETI provisions in November 2000. Although the US claimed that, with the adoption in November 2000 of the FSC Repeal and Extraterritorial Income Exclusion Act (the amended FSC legislation), it had implemented the recommendations and rulings of the DSB, this claim was disputed by the EC, which requested authorization from the DSB to take appropriate countermeasures and suspend concessions pursuant to Article 4.10 of the SCM and Article 22.2 of the Dispute Settlement Understanding (DSU). In response to

<sup>&</sup>lt;sup>25</sup> WTO (1999b), page 23.

an EC request, the WTO established a compliance Panel in January 2001, whose Report was issued in August 2001.

Once again, the WTO Panel ruled against the US, essentially on four grounds. Firstly, it found that the ETI provisions imposed enough special conditions on their use that they were an effective departure from the general US tax practice, and therefore constituted a subsidy (within the meaning of Article 1.1 of the SCM).<sup>26</sup> Secondly, the Panel concluded that the subsidy was "dependent or contingent upon export" performance (within the meaning of Article 3.1(a) of the SCM).<sup>27</sup> Thirdly, the Panel rejected the US argument that the ETI was intended to avoid double-taxation (within the meaning of footnote 59), concluding that the scope of the benefit was considerably broader than the type of income that would ordinarily be at risk of double-taxation.<sup>28</sup> In addition, the Panel found that the 50 per cent "foreign content limitation" violated Article III(4) of GATT 1994 by according less favourable treatment to imported products than to like domestic products.<sup>29</sup> Although the US asked the WTO Appellate Body (AB) to reverse the Panel's findings, the AB issued its Report in January 2002 upholding the Panel's ruling on each of these four main arguments advanced by the US30 Following adoption of both reports by the DSB and failure by the US to take further compliance actions, an arbitration proceeding was carried out pursuant to Article 22.6 of the DSU.

In August 2002, the Arbitrator's award was circulated. The Arbitrator determined that the suspension by the EC of concessions under the GATT 1994 in the form of the imposition of a 100 per cent *ad valorem* charge on imports of certain goods from the US in a maximum amount of US\$4,043 million per year, as described in the EC's request for authorization to take countermeasures and suspend concessions, would

<sup>26</sup> WTO (2001b), page 23.

constitute appropriate countermeasures within the meaning of Article 4.10 of the SCM. In April 2003, the EC requested authorization from the DSB to suspend concessions or other obligations under Article 22.7 of the DSU and Article 4.10 of the SCM. In May 2003, the DSB authorized the EC to take appropriate countermeasures and to suspend concessions. The EC in turn began to impose retaliatory tariffs, as authorized, in a phased manner, beginning in March 2004.

In October 2004, after more than two years of complex negotiations between the House and Senate, the US Congress passed legislation to repeal the FSC/ETI and replace it with a new corporate tax law, which provides for the phasing out of the FSC by 2007 and its replacement with US\$138 billion in tax relief for domestic manufacturing, US multinationals, and a wide range of other industries and businesses. The new legislation called the American Jobs Creation Act of 2004, which some have described as the most significant corporate tax bill since 1986, was signed into law by the President on 22 October 2004.

Shortly afterwards, the EC moved to lift sanctions that it had imposed on US exporters. At the same time, however, the EC expressed concern over transition provisions in the newly-passed legislation repealing the FSC/ETI, one involving the phase-out of the ETI over two years<sup>31</sup>, the other concerning the "grandfathering" of existing contracts so that the full benefit of the ETI applies to exports made under contracts entered into before 17 September 2003. EU officials have argued that the "grandfathering" provision favours exporters of capital goods that have long delivery times; such exporters include Boeing, Microsoft, Intel, Motorola, and Caterpillar. In any event, on 5 November 2004 the EC initiated proceedings at the WTO challenging these provisions. Some United States' observers have suggested that the EC appeal is linked to a separate WTO complaint lodged by the United States against Airbus (see below).

<sup>&</sup>lt;sup>27</sup> Ibid, page 32.

<sup>&</sup>lt;sup>28</sup> Ibid, page 43.

<sup>&</sup>lt;sup>29</sup> One of the conditions of eligibility for tax relief under the ETI measure was that no more than 50 per cent of the fair market value of qualifying property be attributable to the article produced or direct labour performed outside the US.

<sup>&</sup>lt;sup>30</sup> WTO (2002), pages 77-78.

<sup>&</sup>lt;sup>31</sup> This provision permits firms to claim 80 per cent of their otherwise applicable ETI benefit in 2006 and 60 per cent in 2007 before ending in 2008.

### C. CERTAIN INCOME TAX MEASURES USED BY MEMBER STATES OF THE EUROPEAN COMMUNITIES

In an apparent tit-for-tat, the US requested consultations in May 1998 with France, Ireland, Greece, the Netherlands and Belgium in respect of certain income tax measures, which it alleged constitute prohibited subsidies (under Article 3 of the SCM). In the case of France, the measures in question involve the temporary deduction under French income tax law for certain startup expenses of a French company's foreign operations through a tax-deductible reserve account and a special reserve equal to ten percent of the company's receivable position at year end for medium-term credit risks in connection with export sales. The US has contended that each of these measures constitutes an export subsidy, and that the deduction for start-up expenses constitutes an import substitution subsidy. The complaint by the US against Ireland is with regard to "special trading houses", which qualify for a special tax rate in respect of trading income from the export sale of goods manufactured in Ireland. In the case of Greece, the complaint concerns Greek exporters' entitlement under Greek income tax law to a special annual tax deduction calculated as a percentage of export income. As regards the Netherlands, the complaint concerns a provision of Dutch income tax law, which allows exporters to establish a special "export reserve" for income derived from export sales. Finally, the measure in question in Belgium is a special (index linked) tax exemption accorded to Belgian companies for recruiting a departmental head for exports (known as an "export manager"). Consultations are still pending with regard to these measures.

# D. CANADIAN MEASURES AFFECTING THE EXPORT OF CIVILIAN AIRCRAFT

The matter of direct taxation also arose in the dispute between Brazil and Canada over measures affecting the export of civilian aircraft.<sup>32</sup> More specifically, Brazil asserted that Canada's exemption of the Export Development Corporation (EDC) from corporate income tax place the EDC at a decided advantage in the market place. However, this allegation was rejected by the Panel (WTO, 1999a).

### E. United States' Measures Affecting Trade in Large Civil Aircraft

When the US requested consultations with the EC and certain Member States concerning subsidies for large civil aircraft allegedly inconsistent with the Article XXIII:1 of the GATT 1994 and Articles 4, 7 and 30 of the SCM, the EC made its own request on similar grounds.<sup>33</sup> These two disputes each entail larger amounts of assistance than even the FSC/FTI.<sup>34</sup> Unlike the US complaint against the EC and its Member States, which concentrated on government grants, loans and equity infusion, the EC's complaint initially cited several direct tax measures, notably the FSC/ETI and Research and Experimentation Tax Credits.35 both provided by the Federal Government, together with incentives offered by three states (Illinois, Kansas and Washington) and local governments.<sup>36</sup> As discussed earlier, the European Commission has also expressed concern that the transition provisions in the legislation repealing the FSC/ETI scheme would allow companies such as Boeing to continue to benefit from tax relief under the scheme.

<sup>&</sup>lt;sup>32</sup> See WTO (1999a).

<sup>&</sup>lt;sup>33</sup> See WTO documents WT/DS316/1 and WT/DS317/1, 12 October 2004

<sup>&</sup>lt;sup>34</sup> The US alleges that Airbus has received more than US\$15 billion in government assistance since 1967, while the EC claims that Boeing has received US\$23 billion since 1992.

<sup>&</sup>lt;sup>35</sup> Research and Experimentation Tax Credits were mentioned in the EC's request for consultations, but not in its subsequent request for the establishment of a panel (See WTO documents WT/DS317/2, 3 June 2005).

<sup>&</sup>lt;sup>36</sup> The State of Washington, for example, agreed in 2003 to provide a package of some US\$3 billion in tax incentives to Boeing (Watkins, 2004); the incentives include reduced tax rates on income and tax credits (Washington State Department of Revenue, 2004).

### IV. SOME DIRECT TAX MEASURES IDENTIFIED IN TRADE POLICY REVIEWS

In addition to the above measures subject to dispute at the WTO, other kinds of direct tax measures have been identified as instruments of economic policy in the course of Trade Policy Reviews. Such measures, which may or may not be consistent with WTO agreements, can be broadly categorized as: (a) those protecting domestic producers from imports; (b) those providing assistance for exports; (c) those subsidizing domestic production; and (d) those aimed at deterring or attracting foreign investment. Some examples of these types of measures and their use in different countries are found below.

# A. DIRECT TAX MEASURES PROVIDING PROTECTION AGAINST IMPORTS

Direct tax measures may have effects very similar to an import tariff. For example, in some countries, including Member States of the EU, contributions to private pension plans and life insurance policies may qualify for relief under the personal income tax, provided the pension plan or life insurance policy is supplied by domestic companies (who may be foreign-owned).37 In Malaysia, income from annuities purchased from foreign-owned life insurance companies did not qualify for the personal tax exemption otherwise applicable to such income, even if those companies were established in Malaysia.<sup>38</sup> Although governments often argue that such restrictions are for prudential reasons and perhaps to prevent tax fraud, such tax relief is nonetheless equivalent to an import duty because it protects domestic providers of such financial services from providers located abroad.39

In an effort to combat tax evasion, Bangladesh levies a so-called advance income (or withholding) tax at the rate of 2.5 per cent on imports. <sup>40</sup> This tax is creditable for corporate tax purposes, which means that it does not constitute an additional levy on imports as long as corporate (or personal) taxes payable are sufficient to be offset by the tax. However, if the importer is in a non-tax paying position, possibly because the importer is operating at a loss for income tax purposes or is enjoying a corporate tax holiday, the 2.5 per cent levy is tantamount to an import surcharge. <sup>41</sup>

#### B. DIRECT TAX MEASURES AS EXPORT ASSISTANCE

Some direct tax measures may constitute assistance to exports. Indeed, many WTO Members, including Bangladesh, China, India, Malaysia, Papua New Guinea and the Solomon Islands, provide relief from corporate taxes for income from exports or export activities.<sup>42</sup> Relief may take the form of lower corporate tax rates for export-oriented enterprises or adjustments to the tax base (such as a double deduction for certain expenses related to exports). In China, for example, "foreign-invested enterprises" (FIEs) exporting at least 70 per cent of their output qualify for a 50 per cent income tax reduction (or possibly more if they are located in special zones) and a full refund of the income tax paid on the amount of their profits that they reinvest in export-oriented businesses (for at least five consecutive years). The economic effect of such measures is not fundamentally different from that of levying a non-discriminatory corporate income tax combined with an export subsidy (related in a potentially complicated way to the company's income tax situation).

<sup>&</sup>lt;sup>37</sup> In Korea, it has been alleged that the purchase of a foreign automobile can trigger a tax audit, thereby dissuading taxpayers from purchasing imported cars.

 $<sup>^{38}</sup>$  Section 49 of the Income Tax Act 1967 and Section 127 of the Act read with paragraph 6 in Schedule 6 of the Act (WTO, 2001a, p.55).

<sup>&</sup>lt;sup>39</sup> In the EU, such tax relief has been identified as an impediment to labour mobility across Member States. In addition, special corporate tax treatment is often accorded to insurance companies' contributions to policy-reserves used to fund future payments to life insurance policy-holders.

<sup>&</sup>lt;sup>40</sup> A similar tax of 0.25 per cent is also levied on exports.

<sup>&</sup>lt;sup>41</sup> Such a levy would counteract the effects of tax holidays that are widely used in Bangladesh to attract FDI (although the effectiveness of tax holidays is, in any event, rather dubious).

<sup>&</sup>lt;sup>42</sup> Full or partial exemption of direct taxes on income related to exports is included in the illustrative list of export subsidies found in Annex I of the SCM. Article 27.4 of the SCM accorded developing country Members an eight-year transition period for phasing out export subsidies.

### C. DIRECT TAX MEASURES AS ASSISTANCE TO DOMESTIC PRODUCTION AND OTHER ACTIVITIES

Direct tax measures have long been used to assist domestic production of goods and services in certain sectors (such as agriculture, manufacturing and various services) and activities (such as R&D). Among the most common types of direct tax measures are reduced rates of corporate tax, if not complete tax holidays, usually for a prescribed period of time. Reduced rates or tax holidays can apply to specific sectors or activities. Other widely used direct tax measures include accelerated depreciation allowances and investment tax credits, often aimed at specific types of activity. Indeed, specificity in the provision of relief from direct taxes can be achieved with superficially uniform rules by, for example, maintaining a uniform tax rate but manipulating other aspects of the tax base, such as depreciation schedules, or applying investment tax credits. It is widely known that, even if explicitly-specific features of countries' tax laws are ignored, effective taxation of income from capital varies widely across sectors in most Organisation for Economic Cooperation and Development (OECD) countries.<sup>43</sup> Accelerated depreciation for machinery and equipment, for instance, would tend to reduce taxes, and therefore product costs, differentially across sectors depending on their capital intensity. Interestingly, Canada apparently introduced accelerated depreciation for a particular type of capital used in manufacturing in reaction to quite different US tax provisions - the DISC enacted in 1971 - which were subsequently found to be an illegal export subsidy under the GATT.

In China, 40 per cent of investment involving the purchase of domestically-made machinery and equipment can be deducted from enterprise income tax, thereby favouring domestically-produced machinery over imports. Malaysia and Singapore exempt from tax shipping company income derived from the operations of ships owned by domestic companies; this constitutes an incentive for the use of shipping services provided by resident companies.

Direct tax measures are also used to deter or attract foreign investment. A long-standing potential deterrent to foreign investment is the double taxation by source and residence countries of cross-border investment income flows, which bilateral tax treaties are designed to reduce, if not eliminate (see below).

However, direct tax measures, including tax holidays and tax rate reductions, may also be used to encourage foreign investment. In China, for example, "foreign-invested enterprises" are subject to a 15 per cent corporate tax rate for three years following a two-year tax holiday instead of the standard 33 per cent rate applied to purely domestic firms.<sup>44</sup>

Most OECD countries other than the US have negotiated "tax sparing" provisions with their tax treaty partners as a means of ensuring that the relief associated with tax incentives offered by developing countries to foreign investors is not offset by taxation in those investors' country of residence owing to the latter's use of the credit method of relieving for international double

D. Direct tax measures to deter or attract foreign investment

<sup>&</sup>lt;sup>43</sup> See OECD (1991) and Jorgenson and Landau (1993).

<sup>&</sup>lt;sup>44</sup> This discrepancy in tax rates provides an incentive for "round tripping" by Chinese companies, thereby exaggerating the extent of FDI inflows into China.

taxation.<sup>45</sup> In effect, "tax sparing" (and, indeed, the exemption method of providing relief for international double taxation) is a means by which the tax system of the capital-exporting country can be made to accommodate the tax incentives of developing countries (Short, 1966).<sup>46</sup> It results in the capital-exporting country providing a credit for more tax than the investor actually pays in the developing country, and, as such, a subsidy for the export of capital and thus to the production of goods and services in the capital-importing country. Moreover, insofar as the tax

incentives offered by capital-importing developing countries are conditional upon exports or aimed at providing assistance to production in specific sectors, and therefore possibly infringe WTO rules, "tax sparing" in respect of these incentives might also be considered an infringement of these rules.

45 When a developing country offering such incentives

the best practices identified in the OECD (1998) Report, such as having time-limited provisions and anti- abuse clauses.

concludes a treaty with a developed country, there would be a risk, without a tax sparing provision, of these incentives being nullified. Suppose, for instance, that the developing country provides a tax holiday to companies engaged in a certain activity. Dividends, interest or royalties paid by such a company operating in the developing country to residents of the developed country would be taxable in their hands and having paid no tax in the developing country, there would be a correspondingly higher amount of tax to pay in the developed country. In other words, the country of residence of the investor (that is, the developed country) "spares" the tax that it would normally impose on the untaxed (or low-taxed) income earned by its resident abroad by granting foreign tax credits equal to, or possibly greater than, the tax that would otherwise have been paid in the developing country. The Commentary to Articles 23 A and 23 B of the OECD Model discusses tax sparing at paragraphs 72 to 78, in the terms of an OECD (1998) Report on tax sparing. It notes some of the difficulties of tax sparing, such as the possibility for tax abuse and issues of its effectiveness as a form of aid to economic development. While not totally rejecting the use by OECD Members of tax sparing provisions, the conclusion is that tax sparing should only be considered for countries at an economic level considerably below that of OECD Members, should be based on objective economic criteria and should take into account

<sup>&</sup>lt;sup>46</sup> "Tax sparing" is contrary to the principle of capital export neutrality, but consistent with the principle of capital import neutrality (see section 5(c)(iii) below).

# V. RATIONALE AND PRINCIPLES UNDERLYING WTO RULES AND INTERNATIONAL TAXATION

The over-arching goals of the WTO (subject to certain conditions as set forth in the various WTO agreements) and those concerning international taxation appear to be broadly similar, namely the dismantling of barriers to the cross-border flow of goods, services, capital, labour and technology so as to achieve a more efficient allocation of global resources and thereby raise standards of living world-wide. Nonetheless, there are some significant differences between the basic principles underlying the WTO agreements and those applied in international taxation agreements, as well as departures from these principles. This is particularly true of the principles of nondiscrimination and reciprocity. To some extent, these principles reflect the underlying economic rationale for these international agreements. There are also major differences in approach to dispute settlement and the extent to which agreements are binding; WTO agreements are subject to binding dispute settlement procedure, tax treaties are not, although private citizens do have greater rights under tax treaties. These differences are in addition to the fact that whereas WTO rules are, by and large, negotiated multilaterally, international tax rules, with a few exceptions, involve unilateral relief from international double taxation in the form of foreign tax credit or exemption arrangements and a network of some 2000 bilaterally negotiated treaties (mostly based on the OECD Model), which lower withholding tax rates on interest, dividends and royalties. Questions have been raised over the economic rationale underlying multilateral trade agreements and bilateral tax treaties.

# A. THE RATIONALE FOR THE GATT/WTO AGREEMENTS AND TAX TREATIES

One would expect the principles underlying international agreements on trade and taxation to reflect to some extent the economic rationale for such agreements. However, the economic rationale for these agreements is by no means obvious.

It would appear that the WTO agreements and international taxation, notably tax treaties, have broadly similar primary goals. The Marrakesh Agreement (establishing the WTO) states that trade and economic relations "should be

conducted with a view to raising standards of living, ... and expanding the production of and trade in goods and services, while allowing for the optimal use of the world's resources ... by entering into reciprocal and mutually advantageous arrangements directed to the substantial reduction of tariffs and other barriers to trade and to the elimination of discriminatory treatment in international relations". Similarly, the introduction to the OECD's model tax treaty states that the primary goal of treaty formation is "removing the obstacles that double taxation presents", thus reducing its "harmful effects on the exchange of goods and services and movements of capital, technology, and persons" (OECD, 1997, p.I-1). Notwithstanding these broadly similar stated goals, doubts have been expressed as to whether such international agreements are necessary for their achievement.

In the case of the GATT and related WTO agreements, for example, standard economic theory suggests that unilateral free trade is the optimal policy for a national welfare maximising government. Indeed, Krugman (1991), inter alia, has argued that the GATT "makes no sense in terms of economics". However, another body of literature, notably Bagwell and Staiger (2002), maintains that such agreements do, in fact, make economic sense. More specifically, under this so-called traditional economic approach to trade agreements, a government (of a large country) is assumed to set its import tariff, for example, in order to maximize national welfare, while recognizing that some of the burden of the tariff is shifted onto foreign exporters whose products sell at a lower world price (that is, at diminished terms of trade).47 This "terms-of-trade externality" leads governments to set unilateral tariffs that are higher than would be efficient from a global

can be found in Bagwell and Staiger (2002).

\_\_\_

<sup>&</sup>lt;sup>47</sup> Even seemingly small countries may be able to shift the burden of tariffs (and non-tariff barriers) onto foreign exporters insofar as they too can influence the terms of trade where industry is monopolistically competitive or where transportation costs encourage greater trade between proximate countries. A survey of empirical evidence concerning the relevance of terms-of-trade considerations

standpoint.<sup>48</sup> The effect of the GATT and related WTO agreements, therefore, is to eliminate the terms-of-trade driven restrictions in trade volume that arise when policies are set unilaterally, and thereby offer governments a means of escape from a "Prisoners' Dilemma".<sup>49</sup> This rationale for multilateral trade agreements has become the conventional wisdom among international trade theorists.<sup>50</sup>

Similarly, in view of the fact that relief for international double taxation under the credit or exemption systems is currently provided by most countries around the world on a unilateral basis<sup>51</sup>, the question arises as to why treaty relief in the form of the mutual lowering of withholding taxes (on interest, dividends and royalties) needs to be negotiated bilaterally between the "source" and "residence" countries. In the case of the predominant credit system, if unilateral relief already exists in each country, a negotiated

.....

reduction in tax rates on bilateral flows of income from capital may have little, if any, effect on the economic behaviour of foreign investors. Bilaterally negotiated reductions in tax rates will tend more to generate lump sum transfers between governments than to affect the behaviour of foreign investors (Whalley, 2001).52 Indeed, Dagan (2000) maintains that the use of tax treaties to promote FDI is "a myth". Empirical support for this conclusion can be found in a recent study by Blonigen and Davies (2002) using data for OECD countries. Instead, the rationale underlying tax treaties may lie more in preventing "aggressive" tax planning, if not evasion (often by means of deferral, transfer pricing or treaty shopping), which, if not addressed, could result in little, if any, taxes being paid in either the "source" or "residence" country. If investors could avoid taxes altogether, this would tilt the playing field in favour of FDI rather than against it, which would also tend to distort international investment flows. Thus, bilateral tax treaties attempt to strike a balance between the alleviation of international double taxation and the enforcement of single international taxation. Given the wide differences in countries' tax systems, it is argued that the elimination of international double taxation and effective enforcement of single taxation requires bilateralism and therefore some degree of discrimination between countries. As predicted by arguments that treaties are intended more to reduce tax evasion than to promote FDI, Blonigen and Davies (2002) find that under certain circumstances, bilateral tax treaties may actually reduce FDI.

#### B. RECIPROCITY

Broadly speaking, mutual adjustments in trade policy conform to the principle of reciprocity if these policy adjustments result in a roughly similar changes in a country's import and export volumes. In other words, governments seek to achieve a "balance of concessions" in their negotiations (Bagwell and Staiger, 2002). In this way, reciprocity tends to neutralize externalities

the withholding taxes applied to repatriated funds.

<sup>&</sup>lt;sup>48</sup> While the terms-of-trade externality is not the only possible cross-border externality, it is the externality that has figured most prominently in the theoretical literature. Even in the absence of terms-of-trade movements, unilateral tariff choices can be inefficient in the presence of monopolistic competition (Helpman and Krugman, 1989). A "scale externality" may arise if production technologies exhibit increasing returns to scale, in which case the value of the trade agreement can be influenced by the volume of trade between other countries (Ethier, 1998). Ethier (2004) also considers the possibility of "political externalities" based on imperfect or asymmetric information endowing lobbies with distinct abilities to lobby governments about different policies.

<sup>&</sup>lt;sup>49</sup> The Prisoners' Dilemma, attributed to A.W. Tucker, involves a strategic interaction in which two players both gain individually by not cooperating, leading to a Nash equilibrium in which both are worse off than if they cooperated. It explains why countries may choose protection, even though all lose as a result.

<sup>&</sup>lt;sup>50</sup> The terms-of-trade consequences of trade policy can be expressed equivalently in the more familiar language of "market access", which dominates real-world trade negotiations. When a country raises its tariff, the loss in market access that home country exporters experience is simply the "quantity effect" that accompanies the "price effect" of a deterioration in the home countries terms of trade. Hence, the terms-of-trade consequences and the market access implications of trade policy choices are different ways of expressing the same thing (Bagwell and Staiger, 2002).

<sup>&</sup>lt;sup>51</sup> Whereas the more pervasive credit system involves the "residence" country granting a foreign tax credit against domestic taxes for all (or part of) taxes paid in the "source" country, the exemption system involves the "residence" country exempting foreign source income from domestic taxes if taxes have been paid in the "source" country.

<sup>&</sup>lt;sup>52</sup> Hartman (1985) and Sinn (1993) have argued that withholding taxes are irrelevant for expanding multinational firms as it is cheaper for them to expand their overseas affiliates through retained earnings than through repatriated and re-exported funds. This is because retained earnings avoid

transmitted through world prices. Insofar as countries are able to use trade barriers to shift the terms of trade in their favour, multilateral trade liberalization in accordance with the principle of reciprocity is especially important, as such countries will be less inclined to reduce trade barriers if other countries were unwilling to do so as well.

The principle of reciprocity is also a fundamental aspect of bilateral tax treaties. It is manifest in the mutual reduction of source country withholding taxes on income.

#### C. Non-discrimination

Several different concepts of non-discrimination (or "neutrality") underlie international agreements concerning trade and taxation. The main concepts involve most-favoured-nation (MFN) treatment, national treatment (NT), capital import neutrality (CIN) and capital export neutrality (CEN). The MFN and NT obligations are the cornerstones of the GATT/WTO agreements, with the MFN obligation applying to border measures and the NT obligation to internal measures. The principle of NT also underlies international taxation. In addition, international taxation reflects CIN and CEN, each with distinct implications for global efficiency. Opinions differ as to which of the these two latter concepts of non-discrimination is the most appropriate basis for international taxation. with both being found in countries' tax laws. Whereas economists, by and large, prefer CEN, multinational enterprises (MNEs) tend to stress the importance of CIN owing to the importance they attach to non-discrimination at the corporate level.

#### 1. Most-favoured-nation (MFN) principle

In the context of international trade, the MFN principle, as defined in Article I of the GATT and Article II of the GATS, for example, embodies the notion of non-discrimination as far as imports of the same products and services from different sources are concerned, the intent being to avoid trade diversion. One of the main arguments against bilateral or regional trade agreements is that they could divert more trade than they create and thus be detrimental to global efficiency. Although the ultimate goal of multilateral negotiations under the GATT/WTO is presumably to eliminate tariffs as well as non-tariff barriers to trade, in

the meantime, the MFN principle governs the manner in which such taxes on imported goods and services may be used.

By contrast, there is no general MFN obligation in international taxation. Neither the OECD Model, the United Nations Model Tax Convention nor most tax treaties based on these Models contain an MFN article.53 The reason for the absence of such a provision is that tax treaties, unlike WTO agreements are negotiated bilaterally. Such treaties seek not only to alleviate international double taxation but also to achieve an equitable distribution of tax revenues between the "source" and "residence" countries in the face of bilateral capital flows.<sup>54</sup> Hence, they are tailored to the domestic tax systems of the two negotiating countries and the particular investment and wider relation between them.<sup>55</sup> Nonetheless, bilateral tax treaties create the potential for diversion of international capital flows, which would counteract the benefits from any increased capital flows. To the extent that more foreign investment is diverted from one destination to another than is created, such treaties could conceivably contribute to an inefficient world-wide allocation of capital.

18

the long life of most tax treaties, for a form of MFN treatment if the treaty partner gives a more generous deal to future treaty partners. For example, Norway insisted on the insertion of a clause in its tax treaty with Australia that if the latter reduced its rates of withholding tax further in a treaty with another OECD Member, Australia was obliged to notify Norway and to review the relevant provisions in order to provide Norway with the same treatment as that accorded to the third country. This clause was in fact triggered in 2003 when a new protocol to the Australia-United States Double Taxation Convention came into effect, lowering withholding tax rates in certain circumstances (Lennard, 2004). Such clauses are relatively common in one form or the other.

<sup>&</sup>lt;sup>54</sup> The standard division of the income tax base between the two taxing jurisdiction is as follows: the source country is accorded primary (or exclusive) jurisdiction to tax corporate business income, while the residence country is given primary (or exclusive) jurisdiction to tax investment income (such as dividends, interest and royalties). This division is accomplished by the combination of domestic tax law and bilateral tax treaties. The domestic law of the source country typically taxes all business income arising within its jurisdiction.

<sup>&</sup>lt;sup>55</sup> See Lennard (2004) for a more detailed discussion of this and related points.

### 2. National treatment (NT)

The principle of NT (found in Article III of the GATT and Article XVII of the GATS) entails the commitment by a country to treat imported goods and services as well as those produced by foreign-owned or -controlled enterprises operating in its territory no less favourably than goods and services produced by domestically-owned or -controlled enterprises. Thus, whereas border taxes such as tariffs, for example, must be levied in accordance with the MFN notion of non-discrimination alone, internal taxes should be consistent with both NT and MFN. In other words, internal taxation discriminates neither against imports nor among sources of imports.

The principle of NT also underlies the OECD and United Nations Model Tax Conventions and most tax treaties based on these Models. According to Article 24 of the OECD Model, for example, "Nationals of a Contracting State shall not be subjected in the other Contracting State to any taxation or any requirement connected therewith, which is other or more burdensome than the taxation and connected requirements to nationals of that other State in the same circumstances, in particular with respect to residence, are or may be subjected" (italics added for emphasis). However, the scope of NT in tax treaties falls far short of that provided under the GATT/WTO agreements as regards internal direct taxes.<sup>56</sup> Not only is NT usually implemented solely on a bilateral (and reciprocal) basis in tax treaties, the distinction in Article 24 between residents and non-residents permits countries to treat the former differently from the latter.<sup>57</sup> Thus, although bilateral tax treaties are designed to remove tax impediments to international investment, they do so in a non-neutral fashion. As a consequence, they may divert rather than foster international investment. The bilateral nature of tax treaties also tends to confer an unfair bargaining advantage on a large capital-exporting (capital-importing) country in negotiations with a small capital-importing (capital-exporting) country.

### 3. CIN and CEN

International double taxation has long constituted a potentially important barrier to market access, impeding not just to the free movement of goods and services where indirect taxes are concerned, but also capital and other factors in the case of direct taxes. It thus constitutes an impediment to the efficient world-wide allocation of resources. Double taxation arises in connection with indirect taxes insofar as goods and services are taxed in the country where they are produced and also in the country to which they are exported; by and large, border tax adjustment consistent with WTO agreements ensure that this is not the case, particularly as far as value-added taxes are concerned. The interaction between the source and residence countries' direct tax systems may be such that they discriminate not just against inward and outward FDI generally, but also among countries depending on the source or destination of outward and inward FDI, respectively. In other words, Member countries' direct tax systems may diverge from both CIN and CEN, two distinct notions of economic efficiency often used as benchmarks for countries' tax laws. Differences among Members over the appropriateness of their choice of method to alleviate, if not eliminate, double taxation of income from cross-border investment was, as discussed earlier, at the root of the dispute between the EU and the US over the latter's FSC/ETI scheme.

Under CIN, no distinction would be made between residents and non-residents. For CIN to prevail with respect to direct taxation, the total marginal effective tax rate on income from an investment undertaken in a given country (inclusive of taxes in both the source and residence countries) must be the same, irrespective of where the recipient of such income resides. Hence, direct taxation would not influence who invests in a particular country. CIN would ensure that, as far as taxation is concerned, domestic MNEs are not placed at a competitive disadvantage compared to foreign MNEs in markets abroad. Such a situation could arise if (in accordance with CEN) a domestic MNE investing in a relatively low-tax country had to pay tax at the higher domestic rate on the income generated. This could be considered "unfair" to MNEs because it would impair their ability to compete with local firms, or with MNEs based in other countries that levy lower rates of tax on foreign-source income. Since competition among

 $<sup>^{56}</sup>$  Moreover, not all bilateral tax treaties conform with Article 24.

<sup>&</sup>lt;sup>57</sup> Such a distinction is justified by tax authorities on the grounds that cross-border investment and income flows increase the opportunities for avoiding or evading taxes and makes enforcement more difficult, particularly in the case of portfolio investment.

MNEs is desirable on economic efficiency grounds, it follows that Members' tax policies should not distort global investment flows by discouraging the most efficient firms from operating in what would otherwise be their least-cost location. No such tax distortion would arise if CIN were the basis for Member' income tax policy with respect to MNEs. CIN is therefore concerned with "competitive" neutrality. In the absence of any tax discrimination between domestic and foreign firms in the capital-importing country, CIN would be accomplished if income from investment abroad were entirely exempt from taxation in the capital-exporting country. This approach reflects what is known as the "source" (or "territorial") principle of taxation, according to which, national governments tax all income originating within their jurisdictions, regardless of whether that income accrues to residents or to non-residents. It thus embodies the principles of strict NT and MFN treatment as far as taxation in the source (capital-importing) country is concerned, but not with regard to taxation in the residence (capital-exporting) country.

By contrast, the principle of CEN reflects the view that direct taxation ought not to influence the decisions of businesses or individuals residing in a country as to where to invest. CEN is thus concerned with "locational" neutrality and thereby reduces the threat of tax competition among different jurisdictions for FDI. It is accomplished by ensuring that the same total amount of domestic and foreign taxes are paid on an identical investment, irrespective of the country in which that investment is undertaken and the amounts of taxes levied by the foreign country. Under such circumstances, and other things being equal, free mobility of capital would tend to equalize required risk-adjusted pre-tax "hurdle" rates of return<sup>58</sup> on investment across countries, thereby eliminating differences in the cost of capital, and thus international distortions in the demand for capital. If "hurdle rates" of return were equalized across countries, no gain in global output could be accomplished by reallocating capital from one country to another. CEN would be achieved if income were taxed only in investors' country of

58 In general, the "hurdle" (or break-even) rate is defined as the minimum pre-tax rate of return (adjusted for risk and inflation) that is required in order for an investment project to

be profitable (after payment of taxes).

residence, and if there were no discrimination in that country between domestic and foreign-source income. Such a basis for taxation is known as the "residence" (or "world-wide") principle. According to this principle, tax is levied on all income received by domestic residents, regardless of whether that income is from domestic or from foreign sources. It follows that neither NT nor MFN treatment with respect to taxes levied in the source (capital-importing) country are necessary conditions for the achievement of CEN. CEN does require strict NT and MFN treatment with respect to taxes levied by the residence (capital-exporting) country.

Despite the extensive network of bilateral tax treaties designed inter alia to mitigate, if not eliminate, international double taxation, according to evidence reported by the OECD (1991), OECD member countries' tax systems are characterized by a marked lack of both CIN and CEN. The average tax-inclusive hurdle rates for a typical investment in manufacturing undertaken at home by domestic investors tend to be considerably lower than the corresponding rates for outbound and inbound FDI, thus violating the principle Moreover, hurdle rates vary widely of NT. depending on the destination of outbound FDI and the origin of inbound FDI, thereby violating the MFN principle as far as the tax treatment of FDI is concerned and hence possibly diverting FDI from some locations to others. Whether or not such differences in hurdle rates and associated effective tax rates actually influence investors' location decisions is rather uncertain, however.

<sup>&</sup>lt;sup>59</sup> Of course, the estimated hurdle rates for inbound and outbound FDI reported by the OECD (1991), and thus the magnitudes of the departures from CIN and CEN, may be more apparent than real insofar as international investors can circumvent countries' tax laws by means of "creative" accounting and international tax planning, if not outright tax evasion. The effects of taxation on the financing of investment could mitigate those on real investment. Indeed, the reality of the tax laws as they are actually enforced suggests that multinational firms can, and often do, pay less tax than their domestic counterparts. Most analytical models used to measure hurdle (and marginal effective tax) rates, including the one used by the OECD, do not capture the complex interaction between real investment decisions and the various devices that enable multinational firms to shift their tax bases from relatively high- to low-tax countries. The estimated hurdle rates are nonetheless indicative of the potential bias inherent in the design of those tax laws, especially for small and medium-sized firms, which may not have the means to fully exploit international tax planning opportunities.

While taxation may not necessarily be one of the main determinants of investors' decisions regarding FDI, it would be surprising if such differences did not have some influence on those decisions, an influence that is bound to increase as other impediments to FDI are dismantled.<sup>59</sup>

# D. OTHER PRINCIPLES: STABILITY AND TRANSPARENCY<sup>60</sup>

The GATT/WTO agreements provide benefits beyond non-discrimination. Through "bindings", they also impart a degree of stability and predictability to Members' trade policies, thereby fostering international trade. In the case of tariffs, for example, the purpose of "bound" rates is to establish a ceiling on tariff rates that cannot be breached without an offer of compensation to affected trading partners. In addition, the WTO notification process together with periodic reviews of Members' trade policies, practices and measures by the Trade Policies Review Body contribute to the transparency of Members' trade policies.

Tax treaties also offer benefits beyond relief for international double taxation. By establishing a framework for taxation in the source country, foreign investors can make decisions with some confidence about how taxes will continue to be levied in the future. The investment is thus less exposed to dramatic changes in tax policy by the source country, thereby reducing country risk.

### E. DISPUTE SETTLEMENT

GATT/WTO agreements and international taxation have also evolved differently with respect to dispute settlement. The GATT/WTO agreements now provide for binding adjudication. Indeed, as discussed earlier, a considerable number of trade disputes, including a few notable ones involving direct taxation, have been resolved by WTO panels, whose decisions are binding on all parties to the dispute. By contrast, disputes between contracting parties about the application of tax treaties are resolved by mutual agreement between the "competent authorities" of each country. 61 The latter procedure is more diplomatic than legal because these authorities are the tax administrations of the signatory countries. A few bilateral tax treaties (and regional agreements) do provide for arbitration, but binding dispute settlement remains the exception in international taxation (Warren, 2000). However, private citizens do have greater rights under tax treaties than under WTO/GATT agreements. specifically, there is a taxpayer initiation process for the Mutual Agreement Procedure and, perhaps more relevantly, as tax treaties become part of domestic law, taxpayers can invoke the treaty terms as part of domestic law and have a court decision that effectively binds the State.

<sup>&</sup>lt;sup>60</sup> The rules embodied in GATT/WTO agreements and those of international taxation are also designed to address somewhat similar concerns, namely so-called "unfair" trade or competition", in the case of GATT/WTO rules, and "harmful" tax competition, in the case of international tax rules.

<sup>61</sup> See Article 25 of the OECD Model Treaty.

### VI. INCIDENCE OF DIRECT TAXES 62

The terms-of-trade rationale for international trade agreements is closely related to the incidence of tariff and non-tariff measures; that is, the extent to which such measures are passed on as higher prices to domestic consumers. In fact, there is a large body of empirical literature suggesting that tariffs are not fully passed on the domestic consumers, but borne partly by foreign exporters, thereby supporting the terms-of-trade argument.<sup>63</sup>

The dispute over the FSC/ETI (as well as its DISC predecessor) raises a broader empirical issue concerning the incidence of corporate taxes, which is one of the most controversial matters in tax policy analysis. As noted above (in Section 2(c)), whereas WTO rules allow an exported product to be relieved of all indirect taxes levied in the exporting country, thereby allowing border tax adjustments, as in the case of VAT, for example, no such relief is allowed in respect of direct taxes payable by the producing enterprise. The SCM Agreement clarifies that "full or partial exemption, remission or deferral specifically related to exports, of direct taxes or social welfare charges paid or payable"64 by the producing enterprise constitute a prohibited export subsidy. The economic rationale for this rule apparently arises from the assumption that the burden of indirect taxes is generally shifted forward onto the product and is thus reflected in its price, whereas direct taxes are not so shifted, but instead absorbed largely by the owners of the producing enterprise.65 The implication is that whereas indirect taxes have trade effects, direct taxes (or relief therefrom) do not. However, this assumption does not explain SCM rules, where

it is precisely the exemption of export income from direct taxes that is prohibited, presumably because it is believed that such measures may well have trade effects.

The presumption of forward shifting in the case of indirect taxes and backward shifting in the case of direct taxes may be questioned on empirical grounds. Indirect taxes are thought to be fully shifted forward, and therefore affect trade, although the extent of such forward shifting (or "pass through") clearly depends on the demand and supply elasticities of the taxed goods and services (Whalley, 2002). The incidence of capital income taxes is much more controversial. In the case of corporate income tax, on the basis of an econometric study in the US by Kryzaniak and Musgrave (1963), it was thought that during the 1950s the tax was fully shifted forward to consumers in the form of higher product prices, which could have implications for the termsof-trade insofar as the products are exported.66 However, academic thinking on the incidence of the corporate tax during the 1960s and early 1970s came to be dominated by the general equilibrium model of a two-sector closed economy developed by Harberger (1962), who concluded that the tax fell completely on capital, implying no forward shifting and consequently no trade effects of the tax. By the mid-1970s, however, most major studies<sup>67</sup>, which relaxed the closed economy assumption, concluded instead that in an open economy with capital mobile internationally, immobile factors (labour, especially unskilled, and/or land) rather than capital bear most, if not all, of the burden of the tax in the long-run. While there is little debate among economists that

22

<sup>62</sup> The matter of incidence is also relevant to subsidies.

<sup>&</sup>lt;sup>63</sup> See Bagwell and Staiger (2002) for a review of this literature.

<sup>&</sup>lt;sup>64</sup> See item (e) of Annex 1 of the SCM.

<sup>65</sup> For example, in a GATT Working Party on border tax adjustment, most delegations' opinion was that "direct taxes – even assuming that they were partly passed on into prices – were borne by entrepreneurs' profits or personal income." See GATT (1970).

<sup>&</sup>lt;sup>66</sup> Also, by reducing a country's reliance on the international capital market, taxes on capital income have the effect of reducing the present discounted value of future exports and thereby improving the terms of trade (Burgess, 1988).

<sup>&</sup>lt;sup>67</sup> See in particular Bradford (1978), Kotlikoff and Summers (1987), Mutti and Grubert (1985), and Harberger (1995).

capital, being immobile in the short-run, bears most of the burden of the corporate income tax in the short run, most economist now seem to believe that capital escapes most of the burden of the tax in the long run. Indeed, Fuchs, Krueger and Poterba (1997) reported that economists at the top 40 universities estimate, on average that capital bears only 40 per cent of the tax. The question arises, therefore, as to who else bears the burden of such taxes and to what extent.

In a more recent study, using a multi-sector general equilibrium model of two open economies (the domestic economy and the rest of the world) that allows for imperfect substitution between domestic and foreign products as well as between domestic and foreign capital, Gravelle and Smetters (2001) conclude that little of the burden of the corporate income tax is borne by either domestic labour or landowners. Instead, under plausible assumptions involving parameter values, the burden of the tax is borne typically by domestic capital, as argued originally by Harberger (1962). For those parameter values in which the burden is not borne by labour, most of the incidence is instead exported through higher prices<sup>68</sup>; the more a country engages in trade, the more the corporate tax is exported. It follows that the extent to which the burden of the corporate is borne by domestic capital or exported through higher prices depends on the relative importance of trade in countries' economies. Whereas in

large economies, such as the United States, the tax is borne largely by the owners of domestic capital, in smaller open economies the burden is exported. On the other hand, using aggregate consumption Euler equations to describe the capital income tax burden in the US, Mulligan (2004) concludes that capital income taxes seem to be passed on to consumers and workers through higher price mark-ups or lower capital accumulation, or some combination of the two.

Insofar as they are indeed direct taxes, rather than charges related to benefits, payroll taxes are usually viewed in the literature as taxes on labour. The extent to which they affect production costs depends again on the elasticities of demand and supply in the domestic labour market. It is commonly assumed that half these taxes are shifted forward (Whalley, 2002).<sup>69</sup>

Clearly, the incidence of various taxes has a bearing on existing WTO rules concerning tax adjustments at the border in respect of exports. With recent empirical evidence suggesting that the distinction between direct and indirect taxes on businesses for purposes of such adjustment has become rather blurred, a review of WTO rules in the regard might be warranted.<sup>70</sup>

69 As regards, taxes on natural resources, to the extent that

23

-

these involve taxes on pure rent, then the burden of the tax is borne solely by the owners of such resources (Whalley, 2002). In the long-run, however, such taxes may adversely affect new resource exploration activity and thus curtail supply.

<sup>&</sup>lt;sup>70</sup> Hufbauer (2002), for example, has argued that the SCM Agreement should be renegotiated so as to achieve greater parity in rules applied to direct and indirect taxes in the belief that the present distinction between these taxes is artificial. More recently, at the WTO, the United States, in its submission to the Negotiating Group on Rules, has argued that an essential part of the Group's work should involve greater equalization in the treatment of various tax systems that, at least with regard to their subsidy-like effects, have only superficial differences. It believes that the current distinction risks ignoring the potential trade-distorting effect that certain practices involving indirect taxes may have on trade, and may unfairly disadvantage competitors under a direct taxation system (see WTO document TN/RL/W/78, 19 March 2003).

<sup>&</sup>lt;sup>68</sup> This outcome is more likely the higher the substitutability between domestic and foreign investments and the lower the substitutability between domestic and foreign products.

The incidence of taxes also has a bearing on the effectiveness of tax measures in achieving their objectives and thus on the amount of retaliatory measures that ought to be authorized by the WTO in respect of measures found to infringe WTO rules. In the case of the FSC/ETI, for example, countermeasures (in the form of tariffs) amounting to the full value of the subsidy in terms of forgone tax revenue were assessed by the WTO arbitrator, even though the assumption underlying WTO rules is apparently that direct taxes are generally shifted backward rather than forward, which means that relief from such taxes would have little, if any, trade effects. The FSC/ ETI would tend to reduce the before-tax rate of return required of investment in the export sector. thereby attracting investment to the sector and possibly raising US exports. But the magnitude of the rise in exports depends on the extent to which US exporters pass on the associated income tax relief to foreign consumers in the form of lower prices and foreign consumers' responsiveness to these lower prices. Insofar as foreigners do

demand more US exports, this would require them to purchase more US dollars, thereby driving up the exchange rate, which would make those exports more expensive and thereby counteract any export price effect of the FSC/ETI.71 (At the same time, by making imports cheaper, the rise in the exchange rate would tend to increase imports.) It follows that the actual change in US exports resulting from the measure could actually be quite small. Indeed, according to the Congressional Research Office, the increase in the quantity of US exports owing to the FSC ranged from 0.2 to 0.4 of a percentage point; on the basis of 1996 data, the increase in the value of US exports ranging from US\$720 million to US\$1.23 billion, much less than the US\$3 billion cost of the program in forgone tax revenue (Brumbaugh, 2000).<sup>72</sup> With imports estimated to have risen by a similar magnitude, the FSC's effect on net exports is negligible.73

<sup>&</sup>lt;sup>71</sup> Additional upward pressure on the exchange rate would occur to the extent that increased investment in the exporting sector owing to the FSC/ETI came partly from abroad.

<sup>&</sup>lt;sup>72</sup> The Joint Committee on Taxation has estimated that the cost of the ETI in terms of tax revenue forgone was US\$4.8 billion in 2003.

<sup>&</sup>lt;sup>73</sup> Interestingly, these estimates by Brumbaugh (2000) are based on the assumption that <u>all</u> the tax relief associated with the FSC is passed on to foreign consumers as lower prices, which is perhaps unlikely given that some of the main beneficiaries of the FSC, such as Boeing and Microsoft, exercise considerable market power.

### VII. CONCLUSIONS

Hitherto, rules concerning international trade and taxation have, by and large, evolved quite separately. The GATT/WTO agreements are multilateral, requiring MFN as well as national treatment and subject to binding dispute settlement procedures. By contrast, international taxation, if not decided unilaterally, is agreed bilaterally, requiring national treatment and subject to non-binding (diplomatic) dispute settlement. This separate evolution of the two set of rules is rather surprising in view of the many parallels and overlaps between international trade and taxation and the apparent analytical equivalence between cross-border flows of products and flows of factors.

However, the widely-followed ruling by the WTO's Dispute Settlement Body against the US concerning the latter's FSC/ETI scheme, which led to the largest retaliation award ever authorized in a dispute at the WTO, confirmed (if there were ever any doubt) that, generally speaking, direct taxes, like indirect taxes (including tariffs), are subject to the multilateral rules of the WTO, notwithstanding efforts by tax authorities to secure specific exemptions for certain direct tax measures in these agreements. This ruling against the US reconfirmed the traditional distinction under international trade rules between direct and indirect taxes, particularly with respect to how such taxes should be treated under the subsidy and border tax adjustment rules of the WTO. The WTO's ruling against the FSC/ETI

scheme prompted the US Congress finally to pass legislation in late 2004 to repeal the scheme as part of a larger overhaul of the US corporate tax system. The most recent disputes between the US and the EC over assistance to large civil aircraft also encompass direct tax measures. It would not be surprising if other WTO-inconsistent direct tax measures were identified in the future, leading to further disputes among WTO Members. Multilateral WTO rules, which are agreed by consensus, can therefore be expected to continue to be an important factor in how Members' shape their tax policies as they undoubtedly will want to avoid having their tax policies successfully challenged in the WTO.

Finally, the incidence of various taxes has a bearing on existing WTO rules concerning tax adjustments at the border in respect of exports. With recent empirical evidence suggesting that the distinction between direct and indirect taxes on businesses for purposes of such adjustment has become rather blurred, a review of WTO rules in the regard might be warranted. The incidence of taxes also has a bearing on the amount of countermeasures that ought to be authorized by the WTO in respect of tax measures found to contravene WTO rules.

### VIII. REFERENCES

AUERBACH, Alan J. and Laurence J. KOTLIKOFF, 1995, Macroeconomics: An Integrated Approach.

BAGWELL, Kyle and Robert W. STAIGER, 2002, *The Economics of the World Trading System*, The MIT Press, Cambridge, Massachusetts.

BLONIGEN, Bruce A. and Ronald B. DAVIES, 2002, "Do Bilateral Tax Treaties Promote Foreign Direct Investment", NBER Working Paper 8834.

BOND, E. W. and S. E. GUISINGER, "Investment Incentives as Tariff Substitutes: A Comprehensive Measure of Protection", *Review of Economics and Statistics* 67, pp. 91-97.

BRADFORD, David F., 1978, "Factor Prices May Be Constant, But Factor Returns Are Not", *Economics Letters* 1 (3), pp. 199-203.

BRUMBAUGH, David L., 2000, "The Foreign Sales Corporations (FSC) Tax Benefit for Exporting: WTO Issues and an Economic Analysis", Order Code RL30684, Congressional Research Service, The Library of Congress, 11 December.

BRUMBAUGH, David L., 2003, "Export Tax Benefits and the WTO: Foreign Sales Corporations and the Extraterritorial Replacement Provisions", Order Code RS20746, Congressional Research Service, The Library of Congress, 22 April.

BRUMBAUGH, David L, 2004, "A History of the Extraterritorial Income (ETI) and Foreign Sales Corporation (FSC) Export Tax-Benefit Controversy", Congressional Research Service, 9 November.

BURGESS, David F., 1988, "On the relevance of export demand conditions for capital income taxation in open economies", *Canadian Journal of Economics* 21(2), pp. 285-311.

DAGAN, Tsilly, 2000, "The Tax Treaties Myth", New York Journal of International Law and Politics, Summer, pp. 939-996.

DALY, Michael, 1997, "Some Taxing Questions for the Multilateral Agreement on Investment (MAI)", *The World Economy* 20(6), pp. 787-808.

ETHIER, Wilfred J., 1998, "Regionalism in a multilateral world", *Journal of Political Economy* 106, pp. 1214-45.

ETHIER, Wilfred J., 2004, "Trade Policies Based on Political Externalities: An Exploration" (mimeo).

FUCHS, Victor, Alan KRUEGER and James POTERBA, 1997, "Why Do Economists Disagree about Policy? The Role of Beliefs About Parameter Values", NBER Working Paper 6151.

GATT, 1970, "Border Tax Adjustments", Report of the Working Party adopted on 2 December 1970 (L/3464).

GRAVELLE, Jane G. and Kent SMETTERS, 2001, "Who Bears the Burden of the Corporate Tax in an Open Economy?", NBER Working Paper 8280.

HARBERGER, Arnold, 1962, "The Incidence of the Corporation Income Tax", *Journal of Political Economy* 70 (June), pp. 215-240.

HARBERGER, Arnold, 1985, "The ABCs of Corporate Tax Incidence: Insights into the Open Economy Case", in *Tax Policy and Economic Growth*, Washington, D.C.: American Council for Capital Formation.

HARTMAN, David, 1985, "Tax policy and foreign direct investment", *Journal of Public Economics*, pp. 107-121

HELPMAN, Elhanan and Paul R. KRUGMAN, 1989, *Trade Policy and Market Structure*, Cambridge: MIT Press.

HUFBAUER, Gary, 2002, "Testimony Before the House Committee on Ways and Means Hearing on the WTO's Extraterritorial Income Decision", February 27.

JACKSON, John H., 1969, World Trade and the Law of GATT.

JORGENSON, Dale W. and Ralph LANDAU (Editors), 1993, *Tax Reform and the Cost of Capital: An International Comparison*, The Brookings Institution, Washington, D.C.

KOTLIKOFF, Laurence and Laurence H. SUMMERS, 1987, "Tax Incidence" in *Handbook of Public Economics*, edited by Alan J. Auerbach and Martin S. Feldstein, Amsterdam: North Holland.

KRUGMAN, 1991, "Is Bilateralism Bad?" In E. Helpman and A. Razin, eds., *International Trade and Trade Policy*, Cambridge: MIT Press.

KRYZANIAK, M. and R. A. MUSGRAVE, 1963, *The Shifting of the Corporate Income Tax*, Baltimore: Johns Hopkins.

LENNARD, Michael, 2004, "The GATT 1994 and Direct Taxes: Some National Treatment and Related Issues" (mimeo).

MULLIGAN, Casey, B., 2004, "What do Aggregate Consumption Euler Equations say about the Capital Income Tax Burden", NBER Working Paper 10262.

MUNDELL, R. A., 1957, "International Trade and Factor Mobility", *American Economic Review* 47(3), 321-335 (June).

MUTTI, John and Harry GRUBERT, 1985, "The Taxation of Capital Income in an Open Economy: The Importance of Resident Nonresident Tax Treatment", *Journal of Public Economics* 27 (August), pp. 291-309.

OECD, 1991, Taxing Profits in a Global Economy: Domestic and International Issues (Paris).

OECD, 1997, Model tax convention on income and on capital (Paris).

OECD, 1998, Tax Sparing: A Reconsideration (Paris)

SHORT, R. Alan, 1966, "Tax Treaties with Developing Countries", Canadian Tax Journal 14(2).

SINN, Hans-Werner, 1993, "Taxation and the birth of foreign subsidiaries", Trade, Welfare and Economic Policies, Essays in Honor of Murray C. Kemp, H. Heberg and N. Long, editors, University of Michigan Press, Ann Arbor.

SLEMROD, Joel B., 1995, "Free Trade Taxation and Protectionist Taxation", *International Tax and Public Finance*, pp. 471-489.

WARREN, Alvin C., 2000, "Income Tax Discrimination Against International Commerce", *Tax Law Review* 54.

WASHINGTON STATE DEPARTMENT OF REVENUE, 2004, "Aerospace Industry Tax Incentives Overview" (January).

WATKINS, Marylyn P., 2004, "Lost Revenues, Lost Opportunities: Tax Exemptions in Washington State", *Policy Memo*, Economic Opportunity Institute.

WHALLEY, John, 2001, "Puzzles over International Taxation on Cross-Border Flows of Capital Income", NBER Working Paper 8662.

WHALLEY, John, 2002, "Taxes and Trade" (Mimeo).

WTO, 1999a, "Canada - Measures Affecting the Export of Civilian Aircraft" (WT/DS70/R), 14 April.

WTO, 1999b, United States - Tax Treatment for "Foreign Sales Corporations" (WT/DS108/R), 8 October.

WTO, 2001a, Trade Policy Review - Malaysia, (Geneva).

WTO, 2001b, United States – Tax Treatment for "Foreign Sales Corporations" (WT/DS108/RW), 20 August.

WTO, 2002, United States - Tax Treatment for "Foreign Sales Corporations" (WT/DS108/AB/RW), 14 January.

### OTHER WTO DISCUSSION PAPERS

N° 1	"Industrial Tariff Liberalization and the Doha Development Agenda" (February 2003)	)
	By Marc Bacchetta and Bijit Bora	

- N° 2 "Improving the Availability of Trade Finance during Financial Crises" (November 2003) By Marc Auboin and Moritz Meier-Ewert
- N° 3 "Income Volatility in Small and Developing Economies: Export Concentration Matters" (February 2004). By Marion Jansen
- N° 4 "The Role of Export Taxes in the Field of Primary Commodities" (July 2004) By Roberta Piermartini
- N° 5 "The Global Textile and Clothing Industry post the Agreement on Textiles and Clothing" (July 2004) By Hildegunn Kyvik Nordås
- N° 6 "The Trade, Debt and Finance Nexus: at the Cross-roads of Micro- and Macroeconomics" (November 2004) By Marc Auboin
- N° 7 "Selected Issues concerning the Multilateral Trading System" (November 2004) By Rohini Acharya and Michael Daly
- N° 8 "The Changing Landscape of Regional Trade Agreements" (Avril 2005) By Jo-Anne Crawford and Roberto V. Fiorentino